Author's response to reviews

Title: Sudden deterioration due to intratumoral hemorrhage of ependymoma of the fourth ventricle in an infant during flight

Authors:

Ali Mahdavi (mahdavi@yahoo.com)
Nima Baradaran (nimabar@yahoo.com)
Farideh Nejat (nejat@sina.tums.ac.ir)
Mostafa El Khashab (mostkask@yahoo.com)
Maryam Monajjemzadeh (Monaj@yahoo.com)

Version: 2 Date: 11 February 2008

Author's response to reviews: see over
Dear Editor

We prepared this paper and are sending it for your journal. We state that all of us have read this and it has not been published anywhere. As you have wanted the manuscript has been edited with a native in English language and so the format of paper was changed totally according to instructions.

We are looking forward to receive your opinions about this manuscript.

Regards

Ali Mahdavi M.D
Nima Baradaran M.D
Farideh Nejat M.D, MPH
Maryam Monajemzadeh M.D